

University of Huddersfield Repository

Tosh, Graeme, Clifton, Andrew, Mala, Shereen and Bachner, Mick

Physical health care monitoring for people with serious mental illness

Original Citation

Tosh, Graeme, Clifton, Andrew, Mala, Shereen and Bachner, Mick (2010) Physical health care monitoring for people with serious mental illness. Cochrane Database of Systematic Reviews (3). ISSN 1464-780X

This version is available at http://eprints.hud.ac.uk/16804/

The University Repository is a digital collection of the research output of the University, available on Open Access. Copyright and Moral Rights for the items on this site are retained by the individual author and/or other copyright owners. Users may access full items free of charge; copies of full text items generally can be reproduced, displayed or performed and given to third parties in any format or medium for personal research or study, educational or not-for-profit purposes without prior permission or charge, provided:

- The authors, title and full bibliographic details is credited in any copy;
- A hyperlink and/or URL is included for the original metadata page; and
- The content is not changed in any way.

For more information, including our policy and submission procedure, please contact the Repository Team at: E.mailbox@hud.ac.uk.

http://eprints.hud.ac.uk/

Physical health care monitoring for people with serious mental illness (Review)

Tosh G, Clifton A, Mala S, Bachner M



This is a reprint of a Cochrane review, prepared and maintained by The Cochrane Collaboration and published in *The Cochrane Library* 2013, Issue 2

http://www.thecochranelibrary.com



TABLE OF CONTENTS

| HEADER | 1 |
|---|----|
| ABSTRACT | 1 |
| PLAIN LANGUAGE SUMMARY | 2 |
| BACKGROUND | 2 |
| DBJECTIVES | 3 |
| METHODS | 3 |
| RESULTS | 8 |
| DISCUSSION | 9 |
| AUTHORS' CONCLUSIONS | 10 |
| ACKNOWLEDGEMENTS | 10 |
| REFERENCES | 11 |
| CHARACTERISTICS OF STUDIES | 17 |
| DATA AND ANALYSES | 34 |
| ADDITIONAL TABLES | 34 |
| WHAT'S NEW | 34 |
| HISTORY | 34 |
| CONTRIBUTIONS OF AUTHORS | 35 |
| DECLARATIONS OF INTEREST | 35 |
| SOURCES OF SUPPORT | 35 |
| DIFFERENCES BETWEEN PROTOCOL AND REVIEW | 35 |
| NDFX TERMS | 36 |

[Intervention Review]

Physical health care monitoring for people with serious mental illness

Graeme Tosh¹, Andrew Clifton², Shereen Mala³, Mick Bachner⁴

¹General Adult Psychiatry, East Midlands Workforce Deanery, Nottingham, UK. ²School of Health, Community & Education Studies, Northumbria University, Newcastle-upon-Tyne, UK. ³Nottinghamshire Healthcare, Mansfield, Notts, UK. ⁴Rampton Hospital, Nottinghamshire, UK

Contact address: Graeme Tosh, General Adult Psychiatry, East Midlands Workforce Deanery, Nottingham, UK. graemetosh@hotmail.com.

Editorial group: Cochrane Schizophrenia Group.

Publication status and date: Edited (no change to conclusions), published in Issue 2, 2013.

Review content assessed as up-to-date: 18 January 2010.

Citation: Tosh G, Clifton A, Mala S, Bachner M. Physical health care monitoring for people with serious mental illness. *Cochrane Database of Systematic Reviews* 2010, Issue 3. Art. No.: CD008298. DOI: 10.1002/14651858.CD008298.pub2.

Copyright © 2013 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

ABSTRACT

Background

Current guidance suggests that we should monitor the physical health of people with serious mental illness and there has been a significant financial investment over recent years to provide this.

Objectives

To assess the effectiveness of physical health monitoring as a means of reducing morbidity, mortality and reduction in quality of life in people with serious mental illness.

Search methods

We searched the Cochrane Schizophrenia Group Trials Register (October 2009) which is based on regular searches of CINAHL, EMBASE, MEDLINE and PsycINFO.

We updated this search October 2012 and added 61 new trials to the awaiting assessment section.

Selection criteria

All randomised or quasi-randomised clinical trials focusing on physical health monitoring versus standard care or comparing i) self monitoring vs monitoring by health care professional; ii) simple vs complex monitoring; iii) specific vs non-specific checks iv) once only vs regular checks or v) comparison of different guidance.

Data collection and analysis

The authors (GT, AC, SM) independently screened search results and identified three studies as possibly fulfilling the review's criteria. On examination, however, all three were subsequently excluded.

Main results

We did not identify any randomised trials which assessed the effectiveness of physical health monitoring in people with serious mental illness

Authors' conclusions

There is no evidence from randomised trials to support current guidance and practice. Guidance and practice are based on expert consensus, clinical experience and good intentions rather than high quality evidence.

Note: the 61 citations in the awaiting classification section of the review may alter the conclusions of the review once assessed.

PLAIN LANGUAGE SUMMARY

Physical health care monitoring for people with serious mental illness

In recent years there has been an increasing focus on the physical health of people who suffer from mental illness, it has been recognised that these individuals are at greater risk of physical health problems for a variety of reasons. There are now a number of different guidelines telling us how we should monitor physical health in this population. We searched for randomised trials that looked at the effectiveness of physical health monitoring in preventing deterioration of physical health and maintaining quality of life. Despite the amount of guidance available we found no relevant studies. Consequently there is no evidence from randomised controlled trials that physical health monitoring in people with severe mental illness is useful in preventing deterioration in physical health and maintaining quality of life. This, however, does not mean that physical health monitoring does not affect the physical health of people with severe mental illness.

BACKGROUND

Description of the condition

The definition of severe mental illness with the widest consensus is that of the National Institute of Mental Health (NIMH) (Schinnar 1990) and is based on diagnosis, duration and disability (NIMH 1987). People with serious mental illness have conditions such as schizophrenia or bipolar disorder, over a protracted period of time resulting in erosion of functioning in day to day life. A European survey put the total population-based annual prevalence of serious mental illness at approximately two per thousand (Ruggeri 2000). As a consequence of their illness people with serious mental illness have a significantly reduced life expectancy for a variety of reasons including: poor self care, adverse health behaviours (smoking, sedentary lifestyle) and negative effects from psychotropic medication (weight gain, metabolic syndrome) (Robson 2007). In schizophrenia, for example, life expectancy is reduced by around ten years (Newman 1991). A recent publication has shown that people with schizophrenia have a threefold increase in mortality compared with the general population of England and Wales, and that approximately 81% of that increase is from natural causes, especially cardiovascular disease (Brown 2010). There is historical evidence that sufferers from serious mental illness also have increased rates of infectious diseases (including HIV) (Cournos 2005), non-insulin dependent diabetes, respiratory disease and cancer (Dixon 1999, Robson 2007). Despite this, evidence says that there exists a lack of physical health monitoring in people with serious mental illness in both the primary care (Burns 1998) and in the secondary care setting (Paton 2004).

Description of the intervention

Physical health monitoring can take many forms, and these forms are highly divergent and dependent on environmental and socioe-conomic factors. In some instances monitoring is indicated for a specific group of people because of demographic risk factors; one such population are those suffering from serious mental illness (Robson 2007). People with illnesses such as schizophrenia are at greater risk for a number of conditions. This is compounded by the fact that they are less likely to seek medical advice and more likely to be exposed to medications with potentially negative health consequences (Weinmann 2009). People with serious mental illnesses should stand to benefit greatly from a programme of well organised and regular physical health monitoring.

Monitoring differs from promotion in that its principle aim is to obtain information which can then be acted on to treat or prevent a physical health problem. Promotion, on the other hand, is the provision of information and encouragement to people in the hope that they will act to avoid deterioration of current health or development of future health problems. The effects of physical health promotion/advice for people with serious mental illness is evaluated in another future review in this series: Physical health

advice for people with serious mental illness.

Past reviewers have suggested that "essential routine health monitoring [for people with serious mental illness] should include weight, body mass index (BMI) and waist circumference, blood pressure, lipid profiles, screening for insulin resistance and diabetes, dental checks and eye health checks" (Robson 2007). Physical health care monitoring could, therefore, range from the simplest forms of self-monitoring, through to more systematised self-screening, to well-regulated and guideline-directed monitoring of health by health care professionals.

How the intervention might work

Information obtained from physical health monitoring is often the catalyst for more intensive medical input which can be either curative, palliative or preventative. The routine employment of simple and relatively inexpensive physical health monitoring has the potential to identify current, and pre-empt future, health problems. Subsequent action could improve the quality and duration of life for sufferers of serious mental illness. Additional benefits may include a reduction in dependence on medical services. "There are potential savings to be made on prescribing and acute care budgets through prevention or early detection of serious illness in these groups of service users" (DoH 2006).

Why it is important to do this review

In August 2006 the UK's Department of Heath published a commissioning framework (DoH 2006) which, based on examples of current practice (including pilot programmes and expert advice), was intended to provide best practice guidance on the physical health needs of people with severe mental illness. In conjunction with this publication there has been significant investment in 88 English Primary Care Trusts known as 'Spearhead' Trusts to implement the services it suggested. In addition, a raft of guidance around physical heath monitoring in psychiatry has arisen over recent years from organisations such as the Royal College of Psychiatrists (RCPsych 2009; RCPsych 2009a), the National Institute for Clinical Excellence (NICE 2006), Maudsley Prescribing Guidelines (Taylor 2009) and the Serious Mental Illness Physical Health Improvement Profile (White 2009). At no point does the current commissioning framework or guidance documents refer to evidence from randomised control trials or previous reviews. This pathway of identifying a problem, consultation, creation of guidelines and investment to implement the guidance would appear to make good sense. We feel that, at this point, it would be good to assess available evidence of the effects of physical health monitoring.

OBJECTIVES

1. Primary objective

To investigate the effects of the implementation of specific physical healthcare monitoring compared with standard care in a population suffering from severe mental illness.

2. Secondary objective

To compare types and techniques of monitoring.

METHODS

Criteria for considering studies for this review

Types of studies

We considered all relevant randomised controlled trials, and economic evaluations conducted alongside included randomised controlled trials. We excluded quasi-randomised studies, such as those allocating by using alternate days of the week. When we encountered trials described in some way as to suggest or imply that the study was randomised and where the demographic details of each group's participants were similar, we included them and a sensitivity analysis was undertaken to the presence or absence of these data.

Types of participants

We required that a majority of participants should be within the age range 18 to 65 years and suffering from severe mental disorder preferably as defined by National Institute of Mental Health (NIMH 1987) but in the absence of that, from diagnosed illness such as schizophrenia, schizophrenia-like disorders, bipolar disorder, or serious affective disorders. We did not consider substance abuse to be a severe mental disorder in its own right, however we did feel that studies should remain eligible if they dealt with people with dual diagnoses, that is those with severe mental illness plus substance abuse. We did not include studies focusing on dementia, personality disorder and mental retardation as they are not covered by our definition of severe mental disorder.

Types of interventions

- 1. Physical health care monitoring
- 1.1 General physical health care monitoring in addition to standard care: monitoring can be any means of observation, supervision, keeping under review or measuring or testing at intervals. We defined 'physical health' as 'soundness of body' as opposed to the world health organisation's definition of 'health' which includes mental and social well being (WHO 1948).

- 1.2 Focused physical health care monitoring: adherent to specific guidance e.g. as a result of an identified illness (blood sugar in diabetes) or as a result of pharmacological treatment (weight gain with atypical antipsychotic), in addition to standard care.
- 2. Standard care

Care in which physical health monitoring is not specifically emphasised above and beyond care that would be expected for people not suffering from severe mental illness.

3. Variations in delivery

We feel that there may be important studies comparing different types of monitoring delivered in several ways. We are interested in these studies and will endeavour to include them in a relevant comparison.

- 3.1 Differences in who undertakes the monitoring self monitoring vs monitoring by health care professional.
- 3.2 Differences in complexity of monitoring simple routine check/test vs complex check/test.
- 3.3 Differences in focus of checks specific health check vs non-specific health check.
- 3.4 Differences in pattern of checking once only checks vs regular checks.
- 3.5 Differences in guidance followed one set of guidelines vs another.

Types of outcome measures

For the purposes of this review outcomes we divided outcomes into four time periods, i. immediate (within one week) ii. short term (one week to six months) iii. medium term (six months to one year) and, iv. long term (over one year).

Primary outcomes

- 1. Physical health immediate
- 1.1 Failure to identify a disease state and provide appropriate treatment
- 1.2 Failure to effectively manage a known disease state
- 1.3 Failure to act on known risk factors
- 1.4 Unchecked adverse effects of treatment
- 2. Quality of life
- 2.1 Loss of independence
- 2.2 Loss of Activities of Daily Living (ADL) skills
- 2.3 Chronic pain
- 2.4 Immobility
- 2.5 Loss of earnings
- 2.6 Loss of social status
- 2.7 Healthy days

Secondary outcomes

- 1. Physical health periods other than immediate
- 1.1 Failure to identify a disease state and provide appropriate treatment

- 1.2 Failure to improve management of a known disease state
- 1.3 Failure to act on known risk factors
- 1.4 Unchecked side-effects of treatment
- 2. Adverse event
- 2.1 Number of participants with at least one adverse effect.
- 2.2 Clinically important specific adverse effects (cardiac effects, death, movement disorders, prolactin increase and associated effects, weight gain, effects on white blood cell count)
- 2.3 Average endpoint in specific adverse effects
- 2.4 Average change in specific adverse effects
- 2.5 Death natural or suicide
- 3. Service use
- 3.1 Hospital admission
- 3.2 Emergency medical treatment
- 3.3 Use of emergency services
- 4. Financial dependency
- 4.1 Claiming unemployment benefit
- 4.2 Claiming financial assistance because of a physical disability
- 5. Social
- 5.1 Unemployment
- 5.2 Social Isolation as a result of preventable incapacity
- 5.3 Increased burden to caregivers
- 6. Quality of life/satisfaction with treatment
- 6.1 No clinically important change in general quality of life
- 6.2 Average endpoint general quality of life score
- 6.3 Average change in general quality of life score
- 6.4 No clinically important change in general functioning
- 6.5 Average endpoint general functioning score
- 6.6 Average change in general functioning score
- 7. Economic
- 7.1 Increased costs of health care
- 7.2 Days off sick from work
- 7.3 Reduced contribution to society
- 7.4 Family claiming carers' allowance
- 8. Leaving the studies early (any reason, adverse events, inefficacy of treatment)
- 9. Global state
- 9.1 No clinically important change in global state (as defined by individual studies)
- 9.2 Relapse (as defined by the individual studies)
- 10. Mental state (with particular reference to the positive and negative symptoms of schizophrenia)
- 10.1 No clinically important change in general mental state score
- 10.2 Average endpoint general mental score
- 10.3 Average change in general mental state score
- 10.4 No clinically important change in specific symptoms (positive symptoms of schizophrenia, negative symptoms of schizophrenia)
- 10.5 Average endpoint specific symptom score
- 10.6 Average change in specific symptom score

Search methods for identification of studies

Electronic searches

1. Cochrane Schizophrenia Group Trials Register (October 2009) The register was searched using the phrase:[(*physical* or *cardio* or *metabolic* or *weight* or *HIV* or *AIDS* or *Tobacc* or *Smok* or *sex* or *medical* or *dental* or *alcohol* or *oral* or *vision* or *sight*or *hearing* or *nutrition* or *advice* or *monitor* in title of REFERENCES) AND (*education* OR *health promot* OR *preventi* OR *motivate* or *advice* or *monitor* in interventions of STUDY)]

The Cochrane Schizophrenia Group's Trials Register is compiled by systematic searches of major databases, handsearches of relevant journals and conference proceedings (see Group Module). Incoming trials are assigned to relevant existing or new review titles.

2. The Trials Search Co-ordinator, Samantha Roberts, searched the Cochrane Schizophrenia Group Trials Register register (October 2012) using the phrase:

[(*physical* or *cardio* or *metabolic* or *weight* or *HIV* or *AIDS* or *Tobacc* or *Smok* or *sex* or *medical* or *dental* or *alcohol* or *oral* or *vision* or *sight*or *hearing* or *nutrition* or *advice* or *monitor* in title of REFERENCES) AND (*education* OR *health promot* OR *preventi* OR *motivate* or *advice* or *monitor* in interventions of STUDY)]

The Cochrane Schizophrenia Group's Trials Register is compiled by systematic searches of major databases, handsearches of relevant journals and conference proceedings (see Group Module). Incoming trials are assigned to relevant existing or new review titles.

Searching other resources

1. Unsystematic Search

We undertook unsystematic searches of a sample of the component databases (BNI, CINHAL, EMBASE, MEDLINE and Psych-INFO) to determine if any material may have been overlooked. We searched using specific phrases ('physical health', 'monitoring' and 'mental illness') as the searches that create the Cochrane Schizophrenia Group's register of trials are methodology specific. We did not identify any relevant trials.

2. Reference Searching

We inspected the references of all identified studies for other relevant studies.

3. Personal Contact

We contacted the first author of each included for information regarding unpublished trials.

Data collection and analysis

Selection of studies

Authors GT, AC and SM screened the results of the electronic search. MB inspected a random sample of these results, comprising 10% of the total. The principal reviewer GT and co-reviewer AC inspected all abstracts of studies identified through screening and identified potentially relevant reports. Where disagreement occurred we resolved this by discussion, and where there was still doubt, we acquired the full article for further inspection. We then requested the full articles of relevant reports for reassessment and carefully inspected them for a final decision on inclusion (see Criteria for considering studies for this review). In turn GT and AC inspected all full reports and independently decided whether they met inclusion criteria. We were not blinded to the names of the authors, institutions or journal of publication. Where difficulties or disputes arose, we asked author MB for help and if it was impossible to decide, these studies were added to those awaiting assessment and the authors of the papers contacted for clarification.

Data extraction and management

1. Extraction

Authors GT and AC independently extracted data from included studies. Again, we discussed any disagreement, documented our decisions and, if necessary, we contacted the authors of studies for clarification. With remaining problems MB helped clarify issues and we documented our final decisions. We extracted data presented only in graphs and figures whenever possible, we only included it if two reviewers independently had the same result. We made attempts to contact authors through an open-ended request in order to obtain any missing information or for clarification whenever necessary. Where possible, we extracted data relevant to each component centre of multi-centre studies separately.

- 2. Management
- 2.1 Forms

GT and AC extracted data onto standard, simple forms.

2.2 Data from multi-centre trials

Where possible the authors verified independently calculated centre data against original trial reports.

3. Scale-derived data

We included continuous data from rating scales only if:

- a. the psychometric properties of the measuring instrument had been described in a peer-reviewed journal (Marshall 2000); and b. the measuring instrument was not written or modified by one
- of the trialists for that particular trial; and c. the measuring instrument is either i. a self-report or ii. completed by an independent rater or relative (not the therapist).
- 4. Endpoint versus change data

We preferred to use scale endpoint data, which typically cannot have negative values and is easier to interpret from a clinical point of view. Change data are often not ordinal and are very problematic to interpret. If endpoint data were unavailable, we used change data.

5. Skewed data

Continuous data on clinical and social outcomes are often not normally distributed. To avoid the pitfall of applying parametric tests to non-parametric data, we aim to apply the following standards to all data before inclusion: (a) standard deviations and means are reported in the paper or obtainable from the authors; (b) when a scale starts from the finite number zero, the standard deviation, when multiplied by two, is less than the mean (as otherwise the mean is unlikely to be an appropriate measure of the centre of the distribution, (Altman 1996); (c) if a scale starts from a positive value (such as PANSS which can have values from 30 to 210) the calculation described above will be modified to take the scale starting point into account. In these cases skew is present if 2SD>(S-S min), where S is the mean score and S min is the minimum score. Endpoint scores on scales often have a finite start and end point and these rules can be applied. When continuous data are presented on a scale which includes a possibility of negative values (such as change data), it is difficult to tell whether data are skewed or not. We entered skewed data from studies of less than 200 participants in additional tables rather than into an analysis. Skewed data pose less of a problem when looking at means if the sample size is large, and we entered skewed data from large sample sizes into syntheses.

6. Common measure

To facilitate comparison between trials, we intended to convert variables that can be reported in different metrics, such as days in hospital, (mean days per year, per week or per month) to a common metric (e.g. mean days per month).

7. Conversion of continuous to binary

Where possible, efforts were made to convert outcome measures to dichotomous data. This could be done by identifying cut-off points on rating scales and dividing participants accordingly into 'clinically improved' or 'not clinically improved'. It was generally assumed that if there had been a 50% reduction in a scale-derived score such as the Brief Psychiatric Rating Scale (BPRS, Overall 1962) or the Positive and Negative Syndrome Scale (Kay 1986, Kay 1987), this could be considered as a clinically significant response (Leucht 2005, Leucht 2005a). If data based on these thresholds were not available, we used the primary cut-off presented by the original authors.

8. Direction of graphs

Where possible, we entered data in such a way that the area to the left of the line of no effect indicates a favourable outcome for physical health monitoring.

9. Summary of findings table

We anticipate including the following outcomes in a summary of finding table.

- 1. Physical health immediate
- 1.1 Failure to identify a disease state and provide appropriate treatment
- 1.2 Failure to effectively manage a known disease state
- 2. Quality of life

- 2.1 Loss of Activities of Daily Living (ADL) skills
- 3. Adverse event
- 3.1 Clinically important specific adverse effects (cardiac effects, death, movement disorders, prolactin increase and associated effects, weight gain, effects on white blood cell count)
- 3.2 Death natural or suicide
- 4. Economic
- 4.1 Increased costs of health care
- 5. Social
- 5.1 Social Isolation as a result of preventable incapacity

Assessment of risk of bias in included studies

Again working independently, GT,AC and MB assessed risk of bias using the tool described in the Cochrane Collaboration Handbook (Higgins 2008). This tool encourages consideration of how the sequence was generated, how allocation was concealed, the integrity of blinding at outcome, the completeness of outcome data, selective reporting and other biases. We would have excluded studies where allocation was clearly not concealed.

We did not include trials with high risk of bias (defined as at least 3 out of 5 domains were categorised as 'No') in the meta-analysis. If the raters disagreed, the final rating was made by consensus with the involvement of another member of the review group. Where inadequate details of randomisation and other characteristics of trials are provided, we contacted the authors of the studies in order to obtain further information. Non-concurrence in quality assessment was reported.

Measures of treatment effect

1. Binary data

For binary outcomes we calculated a standard estimation of the random-effect risk ratio (RR) and its 95% confidence interval (CI). It has been shown that RR is more intuitive (Boissel 1999) than odds ratios and that odds ratios tend to be interpreted as RR by clinicians (Deeks 2000). Within the Summary of Findings table we assumed for calculation of the low risk groups that the lowest control risk applied to all data. We did the same for the assumption of the highest risk groups. We used the Summary of Findings table to calculate absolute risk reduction for primary outcomes.

- 2. Continuous data
- 2.1 Summary statistic

For continuous outcomes we estimated a fixed-effect mean difference between groups. We preferred not to calculate effect size measures (standardised mean difference SMD). However, in the case of where scales were of such similarity to allow presuming there was a small difference in measurement, we calculated it and, whenever possible, we transformed the effect back to the units of one or more of the specific instruments.

Unit of analysis issues

1. Cluster trials

Studies increasingly employ 'cluster randomisation' (such as randomisation by clinician or practice) but analysis and pooling of clustered data poses problems. Firstly, authors often fail to account for intra class correlation in clustered studies, leading to a 'unit of analysis' error (Divine 1992) whereby p values are spuriously low, confidence intervals unduly narrow and statistical significance overestimated. This causes type I errors (Bland 1997, Gulliford 1999).

Where clustering is not accounted for in primary studies, we presented data in a table, with a (*) symbol to indicate the presence of a probable unit of analysis error. In subsequent versions of this review we will seek to contact first authors of studies to obtain intra class correlation co-efficient of their clustered data and to adjust for this by using accepted methods (Gulliford 1999). Where clustering had been incorporated into the analysis of primary studies, we present these data as if from a non-cluster randomised study, but adjusted for the clustering effect.

We have sought statistical advice and have been advised that the binary data as presented in a report should be divided by a 'design effect'. This is calculated using the mean number of participants per cluster (m) and the intra class correlation co-efficient (ICC) [Design effect = 1+(m-1)*ICC] (Donner 2002). If the ICC was not reported it was assumed to be 0.1 (Ukoumunne 1999).

If cluster studies has been appropriately analysed taking into account intra class correlation co-efficient and relevant data documented in the report, synthesis with other studies would have been possible using the generic inverse variance technique.

2. Cross-over trials

A major concern of cross-over trials is the carryover effect. It occurs if an effect (e.g. pharmacological, physiological or psychological) of the treatment in the first phase is carried over to the second phase. As a consequence on entry to the second phase the participants can differ systematically from their initial state despite a wash-out phase. For the same reason cross-over trials are not appropriate if the condition of interest is unstable (Elbourne 2002). As both effects are very likely in severe mental illness, we only used data of the first phase of cross-over studies.

3. Studies with multiple treatment groups

Where a study involved more than two treatment arms, if relevant, we presented the additional treatment arms in comparisons. Where the additional treatment arms were not relevant, we did not reproduce these data.

Dealing with missing data

1. Overall loss of credibility

At some degree of loss of follow up data must lose credibility (Xia 2009). For any particular outcome should than 50% of data be unaccounted, we did not reproduce these data or use them within analyses. If, however, more than 50% of those in one arm of a

study were lost, but the total loss was less than 50%, we marked such data with '*' to indicate that such a result may well be prone to bias.

2. Binary

In the case where attrition for a binary outcome is between 0 and 50% and where these data were not clearly described, we presented data on a 'once-randomised-always-analyse' basis (an intention to treat analysis). Those lost to follow up were all assumed to have the same rates of negative outcome as those who completed, with the exception of the outcome of death. A sensitivity analysis was undertaken testing how prone the primary outcomes were to change when 'completed' data only were compared to the intention to treat analysis using the above assumption.

3. Continuous

3.1 Attrition

In the case where attrition for a continuous outcome is between 0 and 50% and completer-only data were reported, we have reproduced these.

3.2 Standard deviations

Where there are missing measures of variance for continuous data but exact standard error and confidence interval are available for group means, either 'p' value or 't' value are available for differences in mean, we calculated standard deviation value according to method described in Section 7.7.3 of the Cochrane Handbook (Higgins 2008). If standard deviations were not reported and could not be calculated from available data, we asked authors to supply the data. In the absence of data from authors, the mean standard deviation from other studies was used.

3.3 Last observation carried forward

We anticipated that in some studies the method of last observation carried forward (LOCF) would be employed within the study report. As with all methods of imputation to deal with missing data, LOCF introduces uncertainty about the reliability of the results. Therefore, where LOCF data has been used in the trial, if less than 50% of the data had been assumed, we reproduced these data and indicated that they are the product of LOCF assumptions.

Assessment of heterogeneity

1. Clinical heterogeneity

To judge clinical heterogeneity, we considered all included studies, initially without seeing comparison data. We simply inspected all studies for clearly outlying situations or people which we had not predicted would arise. Should such situations or participant groups arise these will be fully discussed.

2. Methodological heterogeneity

We considered all included studies initially, without seeing comparison data, to judge methodological heterogeneity. We simply inspected all studies for clearly outlying methods which we had not predicted would arise. Should such methodological outliers arise these will be fully discussed.

3. Statistical

3.1 Visual inspection

We visually inspected graphs to investigate the possibility of statistical heterogeneity.

3.2 Employing the I-squared statistic

Heterogeneity between studies was investigated by considering the I-squared method alongside the Chi^2 'p' value. The I^2 provides an estimate of the percentage of inconsistency thought to be due to chance (Higgins 2003). The importance of the observed value of I I^2 depends on i. magnitude and direction of effects and ii. strength of evidence for heterogeneity (e.g. 'p' value from Chi^2 test, or a confidence interval for I²).

I² estimate greater than or equal to 50% accompanied by a statistically significant Chi² statistic, was interpreted as evidence of substantial levels of heterogeneity (Section 9.5.2 - Higgins 2008) and reasons for heterogeneity were explored. If the inconsistency was high and the clear reasons were found, we presented data separately.

Assessment of reporting biases

Reporting biases arise when the dissemination of research findings is influenced by the nature and direction of results (Egger 1997). These are described in Section 10 of the Cochrane Handbook (Higgins 2008). We are aware that funnel plots may be useful in investigating reporting biases but are of limited power to detect small-study effects. We did not use funnel plots for outcomes where there were ten or fewer studies, or where all studies were of similar sizes. In other cases, where funnel plots were possible, we sought statistical advice in their interpretation.

Data synthesis

Where possible we employed a random-effect model for analyses. We understand that there is no closed argument for preference for use of fixed or random-effect models. The random-effect method incorporates an assumption that different studies are estimating different, yet related, intervention effects. According to our hypothesis of an existing variation across studies, to be explored further in the meta-regression analysis despite being cautious that that random-effects methods does put added weight onto the smaller of the studies - we favoured using random-effect model.

Subgroup analysis and investigation of heterogeneity

1. Subgroup analyses

We anticipate no sub-group analyses.

- 2. Investigation of heterogeneity
- 2.1 Unanticipated heterogeneity

Should unanticipated clinical or methodological heterogeneity be obvious we will simply state hypotheses regarding these for future reviews or versions of this review. We do not anticipate undertaking analyses relating to these.

2.2 Anticipated heterogeneity

We are concerned that focused physical health care monitoring may have different effects than a more general approach. We therefore anticipate some heterogeneity for the primary outcomes and propose to summate all data but also present them separately.

Sensitivity analysis

1. Implication of randomisation

We aimed to include trials in a sensitivity analysis if they are described in some way as to imply randomisation. For the primary outcomes we included these studies and if there was no substantive difference when we added the implied randomised studies to those with better description of randomisation, we then employed all data from these studies.

2. Assumptions for lost binary data

Where assumptions had to be made regarding people lost to follow up (see Dealing with missing data) we compared the findings of the primary outcomes where we used our assumption and compared with completer data only. If there was a substantial difference, we reported results and discussed them but continue to employ our assumption.

RESULTS

Description of studies

See: Characteristics of excluded studies; Characteristics of studies awaiting classification.

Results of the search

The initial search of the Cochrane Schizophrenia Group's register of trials in 2009 was a combined search designed to identify studies which would be relevant to this review, and to another sister review on physical health advice for people with serious mental illness (Tosh 2010). The search identified 2382 references (from 1558 studies). After examining the reports, only three were suitable for further examination and all were excluded. Despite the fact that the Cochrane Schizophrenia Group's register of trials is compiled from large comprehensive and systematic searches for trials, we undertook unsystematic searches of a sample of the component databases (BNI, CINHAL, EMBASE, MEDLINE and PsychINFO) to determine if any material may have been overlooked. We searched using specific phrases ('physical health', 'monitoring' and 'mental illness') as the searches that create the Cochrane Schizophrenia Group's register of trials are methodology specific. We did not identify any further relevant trials.

Included studies

No studies met the criteria for this review.

Excluded studies

One trial was excluded, on the basis that it is an ongoing trial in the recruitment stage which is focusing on the monitoring of mental health parameters and not physical health (Jürgens 2008). Another was excluded as it monitored the effects of a pharmacological intervention on a physical health parameter (Lan 2007). The third exclusion was on the basis that the trial compared the effects of different ways of monitoring a specific anxiety symptom, and did not look at a physical health (Rostow 1980). For details please see Characteristics of excluded studies.

1. Awaiting assessment

Sixty one studies are awaiting assessment.

2. Ongoing studies

We are not aware of any ongoing studies.

Risk of bias in included studies

There were no studies that fulfilled the criteria for inclusion. We did not exclude any studies on the grounds of poor methodology.

Allocation

No studies met the criteria for this review.

Blinding

No studies met the criteria for this review.

Incomplete outcome data

No studies met the criteria for this review.

Selective reporting

No studies met the criteria for this review.

Other potential sources of bias

There were no studies that fulfilled the criteria for inclusion. We did not exclude any studies on the grounds of poor methodology.

Effects of interventions

Currently we know of no randomised studies describing the effects of monitoring physical health care of people with serious mental illnesses.

DISCUSSION

Summary of main results

1. No trial-based evidence

Current medical practice in the UK is led by guidance from bodies such as the National Institute of Clinical Excellence (NICE 2006) and The Maudsley Prescribing Guidelines (Taylor 2009) who predominantly base their guidance on little more than anecdotal evidence, consensus of opinion (Marder 2004) and good intentions. The association between schizophrenia and poor physical health is well established (Robson 2007) and, taken at face value, current guidance seems to make sense. Unfortunately history is littered with treatments and policies which 'seemed like a good idea at the time' but which, with the benefit of hindsight, were, at best, ineffective and, at worst, resulted in harm. Extreme examples of well-intentioned treatments could be trepanation for epilepsy (Adams 1856), ice-pick lobotomies for unruly children (El-Hai 2008), or radium water for high blood pressure (JAMA 1925). More contemporary and more subtle is the wide use of oil of evening primrose oil for many ailments when evidence for efficacy is poor (Bayles 2009). This could mean that hopes are raised inappropriately, and, perhaps other more effective treatments avoided. When it comes to mental health policy, the early legislation for the Care Programme Approach in the UK was well intentioned but ultimately imported a largely wasteful and ineffective package of care from the US (Marshall 1996, Marshall 1998) at a time when even those in the US had found it necessary to substantially evolve the approach into a more effective package (Assertive Outreach). In more recent times the evidence to support the view that specialist mental health services such as Dual Diagnosis Teams, Early Intervention or Assertive Outreach Teams are more beneficial than appropriately supported Community Mental Health Teams is not as strong as would have been originally thought (Ley 2000). Care, and the time of people with serious mental illness is too costly to waste on ideas that are not of proven benefit. Vulnerable people with serious mental illness should surely expect that all

aspects of their care has been subject to some degree of evaluation.

Overall completeness and applicability of evidence

No studies met the criteria for this review.

Quality of the evidence

The three studies we obtained for closer inspection were not excluded because of issues of quality. We were just unable to find any studies that were vaguely relevant, regardless of whether they were high or poor quality.

Potential biases in the review process

The search criteria both on the Cochrane Schizophrenia Group Trials Register (October 2009) and on our unsystematic search (see: Searching other resources) should have been robust enough to detect relevant studies. It is possible, however, that we have failed to identify small studies but we think it unlikely that we would have missed large trials. Studies published in languages other than English, and those with equivocal results, are often difficult to find (Egger 1997). Our search was biased by use of English phrases. However, given that the Cochrane Schizophrenia Group's Register covers many languages but is indexed in English we feel that this would not have missed many studies within the register. For example, the search uncovered 101 studies for which the title was only available in Chinese characters. These were checked for relevance by a Chinese speaking colleague (Jun Xia) and none were identified as possibly relevant to this review.

Agreements and disagreements with other studies or reviews

The only other similar systematic review that we are aware of is Bradshaw 2005. This reports on efficacy of healthy living interventions for people with schizophrenia. They too identified no trials of monitoring. We agree with Bradshaw 2005 that rigorous studies are needed.

AUTHORS' CONCLUSIONS

Implications for practice

1. For people with schizophrenia

Due to the lack of evidence for the current guidance based physical health monitoring it is important that people with schizophrenia expect clinicians to explain their intentions clearly. It would seem reasonable that people with schizophrenia are given the choice of whether they want to be monitored in this way or, whether they would want to add to the body of evidence, one way or another, by being part of a well designed trial from which outcomes relevant to their care would be derived.

2. For clinicians

Clinicians should be aware that current guidance on monitoring the physical health for people with serious mental illness is not supported by any evidence from randomised controlled trials. It would seem reasonable that this is explained to people with serious mental illness. It is possible clinicians are expending much effort, time and financial expenditure on monitoring the physical health of people with serious mental illnesses which is unnecessary, intrusive and costly. Clinicians should, therefore, take a much more

critical view of current guidance and attempt to initiate or get involved with any studies which could provide an evidence-base for this practice.

3. For policy makers or managers

Current policies and guidelines are born out of good intentions and "the evidence for such interventions remains uncertain" (NICE 2006). This puts policy makers in a difficult position galvanising consensus rather than evidence. There remains an enduring concern that "L'enfer est plein de bonnes volontés ou désirs" [Hell is full of good intentions or wishes] (St. Bernard of Clairvaux 1150 quoted in Ammer 2003). Policy makers or managers should be better at recommending active research interest in this area.

4. Note: the 61 citations in the awaiting classification section of the review may alter the conclusions of the review once assessed.

Implications for research

1. General

We could not identify any randomised trials that assessed the effectiveness of physical health monitoring in people with serious mental illness which contradicted the view that current guidance and practice is based on good intentions and expert opinion. Basing care on solely evidence from trials is not realistic (Tanenbaum 2005 and Cooper 2003), however, many treatments or approaches that are not well-evaluated are given to people, when it is actually entirely possible to evaluate these approaches. Health care professionals may be doing far more good than they realise - or far more harm. As part of a duty of care, we argue, that 'what could be known, should be known'.

2. Specific

We realise that much thought and care goes into the design of randomised studies. We have, however, also given this issue some consideration and suggest what we think to be a feasible design see Additional tables (Table 1).

ACKNOWLEDGEMENTS

Thanks to Professor Clive Adams, Lindsey Air and Samantha Roberts and the editorial team at the Nottingham University Cochrane Schizophrenia Group for their unwavering support in the writing of this review.

The Cochrane Schizophrenia Group Editorial Base in Nottingham produces and maintains standard text for use in the Methods sections of their reviews. We have used this text as the basis of what appears here and adapted it as required. Thanks also to Jun Xia who examined the Chinese language titles returned by our search.

REFERENCES

References to studies excluded from this review

Jürgens 2008 {published data only}

Jürgens G, Nordentoft M. Effect of genotyping for cyp450 polymorphisms versus intense clinical monitoring on antipsychotic drug treatment. http://www.clinicaltrials.gov 2008.

Lan 2007 {published data only}

Lan T-H, Chiu H-J. The monitor of serum prolactin level and related clinical observations among individuals with schizophrenia spectrum illnesses in a 3 months aripiprazole trial. http://www.clinicaltrials.gov 2007.

Lan 2008 {published data only}

Lan T, Chiu H, Hu T, Huang H, Wu B, Tuan Y. Prolactin level as an indicator to clinical psychopathological severity in aripiprazole treated schizophrenia. *European Neuropsychopharmacology* 2008;**18**:S440.

Rostow 1980 {published data only}

Rostow CD. The effect of self vs. external monitoring and locus of control upon the pacing and general adjustment of psychiatric inpatients. *Behaviour Research and Therapy* 1980;**18**(6):541–8. [MEDLINE: 7447877]

References to studies awaiting assessment

AstraZeneca 2006 {published data only}

AstraZeneca. A 24-week, international, multi-centre, openlabel, flexible-dose,randomised, parallel-group, phase iv study to compare the effect on glucose metabolism of quetiapine, olanzapine, and risperidone in the treatment of patients with schizophrenia. ClinicalStudyResults.org 2006.

Baptista 2010 {published data only}

Baptista TJ, FernÁ_indez E, Carrizo E, FernÁ_indez V, Sandia I, Prieto D, et al.Polimorphisms of the lep- and lepr genes, metabolic profile after prolonged clozapine administration and response to the antidiabetic metformin. Proceedings of the 163rd Annual Meeting of the American Psychiatric Association; 2010 May 22-26; New Orleans, LA. 2010.

Becker 2005 {published data only}

Becker T. Outcome monitoring and outcome management in inpatient psychiatric care (Ergebnis-Monitoring und Ergebnis-Management in der stationären psychiatrischen Versorgung). http://www.controlled-trials.com 2005.

Bobo 2011 {published data only}

Bobo WV, Bonaccorso S, Jayathilake K, Meltzer HY. Prediction of long-term metabolic effects of olanzapine and risperidone treatment from baseline body mass index in schizophrenia and bipolar disorder. *Psychiatry Research* 2011;**189**(2):200–7.

Bushe 2010 {published data only}

Bushe C, Sniadecki J, Bradley AJ, Poole Hoffmann V. Comparison of metabolic and prolactin variables from a sixmonth randomised trial of olanzapine and quetiapine in schizophrenia. *Journal of Psychopharmacology* 2010;**24**(7): 1001–9.

Byerly 2010 {published data only}

Byerly MJ, Nakonezny PA. A pilot study to determine a prolactin threshold that identifies improved sexual functioning when switching from a prolactin-elevating to a prolactin-neutral antipsychotic. *Clinical Schizophrenia and Related Psychoses* 2010;3(4):209–16.

Carmeli 2012 {published data only}

Carmeli C, Knyazeva MG, Cuenod M, Do KQ. Glutathione precursor n-acetyl-cysteine modulates eeg synchronization in schizophrenia patients: A double-blind, randomized, placebo-controlled trial. PLoS ONE 2012; Vol. 7, issue 2.

ChiCTR-TRC-12002273 {published data only}

ChiCTR-TRC-12002273. Effects of paliperidone extended-release tablets (paliperidone er) and olanzapine on metabolic profile, insulin resistance and Ã-cell function in schizophrenic patients: A randomized 12-week study. http://www.chictr.org/en/proj/show.aspx?proj=3040 2012.

Cordes 2011 {published data only}

Cordes J, Kahl K, Janner M, Muller H, Wagner M, Maier W, et al. Prevalence of the metabolic syndrome in men and women at risk of psychosis. *European Archives of Psychiatry and Clinical Neuroscience* 2011;**261**:S33.

Fernandez 2010 {published data only}

Fernandez E, Carrizo E, Fernandez V, Connell L, Sandia I, Prieto D, et al. Polymorphisms of the LEP- and LEPR genes, metabolic profile after prolonged clozapine administration and response to the antidiabetic metformin. *Schizophrenia Research* 2010;**121**(1-3):213–7.

Henderson 2011 {published data only}

Henderson DC, Freudenreich O, Borba CPC, Wang X, Copeland PM, Macklin E, et al. Effects of modafinil on weight, glucose and lipid metabolism in clozapine-treated patients with schizophrenia. *Schizophrenia Research* 2011; **130**(1-3):53–6.

IRCT201107187049N1 {published data only}

IRCT201107187049N1. An assessment of effects of melatonin on metabolic effects of olanzapine in patients with schizophrenia. http://www.irct.ir/searchresult.php?id=7049&number=1 2011.

ISRCTN63382258 {published data only}

ISRCTN63382258. Early intervention dental trial. http://www.controlled-trials.com 2011.

Jones H, Adams C, Simpson J, Clifton A, Callaghan P, Liddle P. The three shires early intervention dental trial: a real world cluster randomised controlled trial. Proceedings of the East Midlands & South Yorkshire MHRN Hub Annual Research Meeting; 2012 Mar 9; Nottingham, UK 2012

UKCRN11432. The three shires early intervention dental trial. http://public.ukcrn.org.uk/ 2012.

Kent 2011 {published data only}

Kent JM, Daly EJ, Janssens L, Newcomer JW, Van Osselaer N, Husken G, et al.Metabolic and body mass parameters observed with jnj-37822681, a novel fast-dissociating d receptor antagonist, versus olanzapine. *Biological Psychiatry* 2011;**69**(9 Suppl):228. [MEDLINE: BIOSIS: PREV201100354601]

Kluge 2012 {published data only}

Kluge M, Himmerich H, Wehmeier PM, Rummel-Kluge C, Dalal M, Hinze-Selch D, et al. Sleep propensity at daytime as assessed by multiple sleep latency tests (mslt) in patients with schizophrenia increases with clozapine and olanzapine. *Schizophrenia Research* 2012;**135**(1-3):123–7.

Krakowski 2011 {published data only}

Krakowski M, Czobor P. Cholesterol and cognition in schizophrenia: A double-blind study of patients randomized to clozapine, olanzapine and haloperidol. *Schizophrenia Research* 2011;**130**(1-3):27–33.

Kreyenbuhl 2011 {published data only}

Kreyenbuhl J. A patient centered health technology intervention to improve screening for the metabolic side effects of second generation antipsychotic medications. Proceedings of the 164th annual general meeting of the American Psychiatric Association; 2011 May 14-18; Honolulu, Hawaii. 2011.

L'Italien 2006 {published data only}

L'Italien GJ, Blonde L, Hanssens L, McQuade R. Prevalence of diabetes/cardiovascular risk factors among schizophrenia patients treated with atypical antipsychotics in a multicenter, randomized, naturalistic, open-label study (Schizophrenia Trial of Aripiprazole: STAR study). *Diabetes* 2006;55 (Suppl 1):A538.

Lencz 2010 {published data only}

Lencz T, Robinson DG, Napolitano B, Sevy S, Kane JM, Goldman D, et al.Drd2 promoter region variation predicts antipsychotic-induced weight gain in first episode schizophrenia. *Pharmacogenetics and Genomics* 2010;**20**(9): 569–72.

Lieberman 2010 {published data only}

Lieberman J, Jarskog LF, Hamer R, Catellier D, Stewart D, LaVange L, et al. Metformin for obesity and metabolic abnormalities in schizophrenia. Proceedings of the 49th Annual meeting of the Americam College of Neuropsychopharmacology; 2010 Dec 5-9; Miami, Florida. 2010.

Loza 2011 {published data only}

Loza B, Polikowska M, Patejuk-Mazurek I, Bednarski P, Mosiolek A, Wojcik R, et al.Olanzapine standard and orally-disintegrating tablets: A crossover study of metabolic measures. *European Neuropsychopharmacology* 2011;**21**(Suppl 3):S503. [MEDLINE: BIOSIS: PREV201200044578]

Miceli 2010 {published data only}

Miceli JJ, Tensfeldt TG, Shiovitz T, Anziano R, O'Gorman C, Harrigan RH. Effects of oral ziprasidone and oral haloperidol on qtc interval in patients with schizophrenia or schizoaffective disorder. *Pharmacotherapy* 2010;**30**(2): 127–35. [MEDLINE: 20099987]

NCT00418171 {published data only}

Kane JM, Strom BL, Eng SM, Faich G, Reynolds RF, D'Agostino RB, et al. Primary and readjudication mortality results from zodiac, a large simple trial of ziprasidone vs. olanzapine in patients with schizophrenia. *Schizophrenia Research* 2010;**117**(2-3):183–4.

Strom BL, Eng SM, Faich G, Reynolds RF, D'Agostino RB, Ruskin J, et al. The Ziprasidone Observational Study of Cardiac outcomes (ZODIAC): findings from a large simple trial of ziprasidone vs. Olanzapine in real-world use among 18154 patients with schizophrenia. *Schizophrenia Research* 2010;**117**(2-3):311.

NCT00484302 {published data only}

Hjorthaj CR. Validity of self-reported cannabis use as measured by the timeline follow-back instrument in a trial randomizing people with comorbid cannabis use disorder and schizophrenia spectrum disorder. *Early Intervention in Psychiatry* 2010;4(Suppl 1):160.

Hjorthoj CR, Fohlmann A, Larsen AM, Arendt M, Nordentoft M. Correlations and agreement between delta-9-tetrahydrocannabinol (thc) in blood plasma and timeline follow-back (tlfb)-assisted self-reported use of cannabis of patients with cannabis use disorder and psychotic illness attending the capopus randomized clinical trial. *Addiction* 2011;**107**(6):1123–31. [MEDLINE: 22151583]

NCT01075295 {published data only}

NCT01075295. Prevention of weight gain in early schizophrenia and schizoaffective disorder. http://www.clinicaltrials.gov 2010.

NCT01368406 {published data only}

NCT01368406. Lifestyle intervention for weight gain management for patients with schizophrenia. http://ClinicalTrials.gov/show/NCT01368406 2011.

NCT01567124 {published data only}

NCT01567124. Alleviating the metabolic side effects of antipsychotic medications. http://ClinicalTrials.gov/show/NCT01567124 2012.

NCT01606436 {published data only}

NCT01606436. A study measuring effect of pomaglumetad methionil (ly2140023) on electrocardiographs in participants with schizophrenia. http://ClinicalTrials.gov/show/NCT01606436 2012.

Newcomer 2010 {published data only}

Newcomer JW, Nicol GE, Haupt DW, Flavin KS, Schweiger JA, Yingling MD. Changes in adiposity, insulin sensitivity and related metabolic measures during randomized medication switches to aripiprazole from other atypical antipsychotics. *Diabetes* 2010;**59**(Suppl 1):488. [MEDLINE: BIOSIS:PREV201000488139]

Nicol 2011 {published data only}

Nicol GE, Yingling MD, Flavin KS, Schweiger JA, Newcomer JW. Cardiometabolic risk indicators in children before and after 3 months of initial antipsychotic exposure. *Schizophrenia Bulletin* 2011;1:28.

Nielsen 2012 {published data only}

Nielsen J, Thode D, Stenager E, Andersen KO, Sondrup U, Hansen TN, et al. Hematological clozapine monitoring with a point-of-care device: A randomized cross-over trial. European Neuropsychopharmacology 2012;22(6):401–5. [MEDLINE: 22137859]

Ozguven 2011 {published data only}

Ozguven HD, Baskak B, Oner O, Atbasoglu C. Metabolic effects of olanzapine and quetiapine: A sixweek randomized, single blind, controlled study. *Open Neuropsychopharmacology Journal* 2011;4(1):10–7.

Peuskens 2007 b {published data only}

De Hert M, Mittoux A, He Y, Peuskens J. Metabolic parameters in a subset of patients in the scop study. Schizophrenia Research 2010;117(2-3):500–1. De Hert M, Mittoux A, He Y, Peuskens J. Metabolic parameters in the short- and long-term treatment of schizophrenia with sertindole or risperidone. European Archives of Psychiatry & Clinical Neuroscience 2011;261(4): 231–9. [MEDLINE: 20820795]

Saddichha 2011 {published data only}

Saddichha S, Akhtar S. Hypertension, diabetes and cardiometabolic changes associated with antipsychotic use in schizophrenia. *International Clinical Psychopharmacology* 2011;**26**:143.

Scheewe 2011 {published data only}

Scheewe T, Van Haren N, Sarkisyan G, Takken T, Hulshoff Pol H, Backx F, et al. Effect of cardiovascular exercise on global brain volumes in patients with schizophrenia and matched healthy controls. *European Archives of Psychiatry and Clinical Neuroscience* 2011;**261**:S57–S8.

Stroup 2011 {published data only}

Stroup TS, Hamer RM, Ray N, Esscok SM, Lieberman JA. The effect of switching from olanzapine, quetiapine, or risperidone to aripiprazole on risk of cardiovascular disease: Results from the comparison of antipsychotics for metabolic problems (camp) study. Proceedings of the 50th Annual Meeting of the American College of Neuropsychopharmacology; 2011 Dec 4-8;a, Waikoloa, Hawaii. 2011:S101.

Tanasiewicz 2011 {published data only}

Tanasiewicz M, Skucha-Nowak M, Greñ A, Hese RT, Twardawa H, Gorczyca P. Analysis of state of the oral cavity in schizophrenic inpatients treated with atypical and classical neuroleptics [Analiza stanu jamy ustnej chorych na schizofreniê leczonych psychiatrycznie w warunkach szpitalnych neuroleptykami atypowymi i klasycznymi]. *Advances in Clinical and Experimental Medicine* 2011;**20**(2): 187–98.

Tessier 2010 {published data only}

Tessier C, Hoeben D, Bergmans P, Korcsog P, Niehaus D, Ucok A, et al.A comparison of the metabolic effects of paliperidone er vs. Oral olanzapine in recently diagnosed and chronic patients with schizophrenia - results of a randomized controlled trial. *Early Intervention in Psychiatry* 2010;4(Suppl 1):137.

Tiwari 2010 {published data only}

Tiwari AK, Zai CC, Meltzer HY, Lieberman JA, Muller DJ, Kennedy JL. Association study of polymorphisms in insulin induced gene 2 (insig2) with antipsychotic-induced weight gain in european and african-american schizophrenia patients. *Human Psychopharmacology* 2010;**25**(3):253–9. [MEDLINE: 20373477]

何淑芬 2011 {published data only}

何淑芬, 江桂芳, 罗廷芳. Health education path in patients with schizophrenia [健康教育路径在精神分裂症患者中的应用]. 当代护士(专科版) 2011;7:166—7.

侯雪玲 2009 {published data only}

侯雪琦, 钟代曲, 李训军. Health education to improve the initial schizophrenia drug compliance of patients [Google Translate] [健康教育对改善初次住院精神分裂症 患者服药依从性的探讨]. Journal of Qiqihar Medical College [Qiqihar Yixueyuan xuebao][齊齊哈爾醫學院等 羁] 2009; Vol. 30, issue 2:213—4.

冯亚芬 2011 {published data only}

冯亚芬, 郑玉红. System of health education on patients with schizophrenia medication compliance [系 统健康教育对精神分裂症患者服药依从性的影响]. 中国民康医学 2011;23(16):2075—80.

刘根 2010 {published data only}

刘根凤,陈姬,徐萍,王冬云. Systematic health education of diabetic patients with schizophrenia [系统化健康教育对精神分裂症件糖尿病患者的影响]. 齐鲁护理杂志 2010;**16**(25):9—10.

刘清连 2009 {published data only}

刘清连. Health education on compliance of schizophrenic patients of] Google Translate [健康教育对精神分裂症患者治疗依从性的影响]. 中国当代医药 2009;**16**(14):117.

张敏 2010 {published data only}

涨敏, 江志萍. Health education of patients with schizophrenia path of compliance behavior [Google Translate] [健康教育路径对精神分裂症患者遵医行为的影响]. Shanghai Archives of Psychiatry [上海精神医学] [上海精神醫學] 2010; Vol. 22, issue 3:151-3, 188.

张炜 2009 {published data only}

张炜, 钟耕坤, 夏晓. health education on the rehabilitation of schizophrenic patients convalescent Effectiveness [Google Translate] [健康教育对恢复期精神分裂症患者康复的有效性研究]. Journal of Qilu Nursing |济鲁护理杂志/2009;15(23):11-3.

彭燕 2009 {published data only}

彭燕, 王旸, 胡传芬. Analysis of the intervention eficacy of health education for patients with schizophrenia in rehabilitation stage [健康教育对康复期精神分裂症患者 干预效果分析]. *Journal of Qilu Nursing |* 齐鲁护理杂志/ 2009;**15**(9):12-3.

徐清芝 2010 {published data only}

徐清芝,姜芙,赵万,赵志刚,王旸. Health education to improve community-based rehabilitation of schizophrenia symptoms and social function [Google Translate] [健康教育对改善社区康复期精神分裂症患者症状和社会功能的研究]. Chinese Journal of Health Education 2010;26(5):384—6.

方崇芳 2011 {published data only}

方崇芳, 杨世平. The impact of health education on the rehabilitation of patients with schizophrenia [健康教育对康复期精神分裂症患者的影响]. 中国民康医学 2011;23(13):1679-80.

李玉凤 2010 {published data only}

李玉凤,李春燕,王继禹,吴建杰. Interactive health education on the rehabilitation of patients with schizophrenia [Google Translate] [互动式健康教育对精神分裂症患者康复的影响]. Medical Journal of Chinese People's Health |中国民康医学] 2010;22(5):622—3.

李红 2010 {published data only}

李红, 裘瑪珊, 钟进. Observation on effect of the drug treatment of health education for patients with schizophrenia [精神分裂症患者药物治疗健康教育的效果观察]. 矛士走修杂志 2010;**25**(13):1169—71.

李萍 2010 {published data only}

李萍, 邓美华, 陈凤莲, 吴春秀, 罗群. systemic health education on the quality of life of schizophrenic patients] Google Translate [系统性健康教育对精神分裂症病人生活质量的影响 []. 全科护理 2010;8(3B):659-60.

林素兰 2010 {published data only}

林素兰,张夹秋,林贵兰. health education intervention on the impact of schizophrenia patients with constipation] Google Translate [健康教育干预对精神分裂症病人便秘影响的研究 []. 全科护理 2010; Vol. 8, issue 7C:1891—3.

温广妹 2009 {published data only}

鴻广妹、何燕丰、王美、王春梅。 of health education on schizophrenia medication compliance of patients] Google Translate [健康教育 对精神分裂症患者服药依从性的影响 []. *Today Nurse* /当代护士/2009:6:98−9.

王宪琴 2010 {published data only}

王宗琴, 张程赪. full of health education on prognosis of 120 cases were of patients with schizophrenia] Google Translate [全程健康教育对120例精神分裂症患者预后影响对照研究 []. Medical Journal of Chinese People's Health |中国民康医学] 2010;**22**(5):620—1.

王继红 2010 {published data only}

王继红,宋新勤,李莹,曹红军,贾金鼎. health education to improve compliance of patients with schizophrenia survey] Google Translate [健康教育对精神分裂症患者遵医行为改善的调查 []. 中国实用医药 2010;5(21):246—7.

穩世铭 2010 {published data only}

穆世铭,简天容. self-management of health education on compliance of schizophrenic patients of] Google Translate [自我管理健康教育对精神分裂症患者治疗依从性的影响]]. Journal of Chinese Rural Physician [中国社区医师: 综合版] 2010;21:72.

苏勉 2010 {published data only}

苏勉, 苏天勋, 梁艳, 叶健. Regular health education on the families of schizophrenic patients the impact of compliance] Google Translate [家屬定期健康教育对精神分裂症患者治疗依从性的影响 []. Nanfang Journal of Nursing |南方护理学报/2010;17(8B):74—6.

苏爱东 2010 {published data only}

苏爱东, 李军花, 李晓芳. Health education combined antipsychotic efficacy of treatment and rehabilitation of schizophrenia] Google Translate [健康教育合并抗精神病药物治疗 原复期精神分裂症疗效分析 []. Medical Journal of Chinese People's Health |中国民康医学 | 2010;22(4):484.

蔡连秀 2010 {published data only}

蔡连秀. Interactive health education to improve medication compliance in schizophrenia clinical study] Google Translate [互动式健康教育提高精神分裂症患者服药依从性的临床研究 []. Journal of the Gannan Medical College [議南医学院学报] 2010;30(2):239—40.

爾瑞云 2010 {published data only}

爾瑞云. Group health education modules in the application of schizophrenia patients] Google Translate [集体健康教育模块在精神分裂症病人中的应用 []. 全科护理 2010;8(7B):1798—9.

Additional references

Adams 1856

Adams F. The Book of Cures In: The extant works of Aretaeus, the Cappadocian. 1st Edition. London: The Sydenham Society, 1856.

Altman 1996

Altman DG, Bland JM. Detecting skewness from summary information. *BMJ* 1996;**313**(7066):1200.

Ammer 2003

Ammer C. The American Heritage Dictionary of Idioms. http://www.yourdictionary.com/about/The-American-Heritage-Dictionary-of-Idioms.html (accessed 23 December 2009).

Bayles 2009

Bayles B, Usatine R. Evening primrose oil. *American Family Physician* 2009;**80**(12):1405–8.

Bland 1997

Bland JM. Statistics notes. Trials randomised in clusters. *BMJ* 1997;**315**:600.

Boissel 1999

Boissel JP, Cucherat M, Li W, Chatellier G, Gueyffier F, Buyse M, Boutitie F, Nony P, Haugh M, Mignot G. The problem of therapeutic efficacy indices. 3. Comparison of the indices and their use [Apercu sur la problematique des indices d'efficacite therapeutique, 3: comparaison des indices et utilisation. Groupe d'Etude des Indices D'efficacite.]. *Therapie* 1999;**54**(4):405–11. [PUBMED: 10667106]

Bradshaw 2005

Bradshaw T, Lovell K, Harris N. Healthy living interventions and schizophrenia: a systematic review. *Journal of Advanced Nursing* 2005;**49**(6):634–54.

Brown 2010

Steve Brown, Miranda Kim, Clemence Mitchell, Hazel Inskip. Twenty-five year mortality of a community cohort with schizophrenia. *The British Journal of Psychiatry* 2010; **196**:116–121.

BSDH 2000

British society for disability and oral health (Working Group). Oral health care for people with mental health problems - guidelines and recommendations. http://www.bsdh.org.uk/guidelines/mental.pdf (accessed 23 December 2009):1–20.

Burns 1998

Burns T, Cohen A. Item-of-service payments for general practitioner care of severely mentally ill persons: does the money matter?. *British Journal of General Practice* 1998;**48** (432):1415–6. [PUBMED: 9800401]

Cooper 2003

Cooper B. Evidence-based mental health policy: a critical appraisal. *British Journal of Psychiatry* 2003;**183**:105–13.

Cournos 2005

Cournos F, McKinnon K, Sullivan G. Schizophrenia and comorbid human immunodeficiency virus or hepatitis C virus. *Journal of clinical psychiatry* 2005;**66**(Suppl 6):27–33. [PUBMED: 16107181]

Deeks 2000

Deeks J. Issues in the selection for meta-analyses of binary data. Proceedings of the 8th International Cochrane Colloquium; 2000 Oct 25-28; Cape town. Cape Town: The Cochrane Collaboration, 2000.

Divine 1992

Divine GW, Brown JT, Frazier LM. The unit of analysis error in studies about physicians' patient care behavior. *Journal of General Internal Medicine* 1992;7(6):623–9.

Divon 1999

Dixon L, Postrado L, Delahanty J, Fischer PJ, Lehman A. The association of medical comorbidity in schizophrenia with poor physical and mental health. *Journal of Nervous and Mental Disease* 1999;**187**(8):496–502. [PUBMED: 10463067]

DoH 2006

Department of Health. Choosing Health: Supporting the physical needs of people with severe mental illness. www.dh.gov.uk/prod_consum_dh/groups/dh_digitalassets/@dh/@en/documents/digitalasset/dh_4138290.pdf (accessed 5 November 2009).

Donner 2002

Donner A, Klar N. Issues in the meta-analysis of cluster randomized trials. *Statistics in Medicine* 2002;**21**:2971–80.

Egger 1997

Egger M, Davey-Smith G, Schneider M, Minder CSO. Bias in meta-analysis detected by a simple, graphical test. *BMJ* 1997;**13**:629–34.

El-Hai 2008

El-Hai J. Lessons of the first era of psychosurgery. *Clinical neurosurgery* 2008;**55**:138–9. [PUBMED: 19248679]

Elbourne 2002

Elbourne D, Altman DG, Higgins JPT, Curtina F, Worthingtond HV, Vaile A. Meta-analyses involving cross-over trials: methodological issues. *International Journal of Epidemiology* 2002;**31**(1):140–9.

Gulliford 1999

Gulliford MC. Components of variance and intraclass correlations for the design of community-based surveys and intervention studies: data from the Health Survey for England 1994. *American Journal of Epidemiology* 1999;**149**: 876–83.

Higgins 2003

Higgins JP, Thompson SG, Deeks JJ, Altman DG. Measuring inconsistency in meta-analyses. *BMJ* 2003;**327**: 557–60.

Higgins 2008

Higgins JPT, Green S (editors). Cochrane Handbook for systematic reviews of interventions 5.0.1[updated September 2008] The Cochrane Collaboration, 2008. Available from www.cochrane-handbook.org.

JAMA 1925

Beaurea of Investigation, Journal of the American Medical Association. Radium ore revigorator: Capitalizing the public's ignorance of radium and radioactivity. *Journal of the American Medical Association* 925;**85**:1658–60.

Kay 1986

Kay SR, Opler LA, Fiszbein A. *Positive and negative syndrome scale (PANSS) manual.* North Tonawanda, NY: Multi-Health Systems, 1986.

Kay 1987

Kay SR, Fiszbein A, Opler LA. The positive and negative syndrome scale (PANSS) for schizophrenia. *Schizophrenia bulletin* 1987;**13**(2):261–76. [PUBMED: 3616518]

Leucht 2005

Leucht S, Kane JM, Kissling W, Hamann J, Etschel E, Engel R. Clinical implications of Brief Psychiatric Rating Scale scores. *British Journal of Psychiatry* 2005;**187**:366–71. [PUBMED: 16199797]

Leucht 2005a

Leucht S, Kane JM, Kissling W, Hamann J, Etschel E, Engel RR. What does the PANSS mean?. *Schizophrenia Research* 2005;**79**(2-3):231–8. [PUBMED: 15982856]

Ley 2000

Ley A, Jeffery DP, McLaren S, Siegfried N. Treatment programmes for people with both severe mental illness and substance misuse. *Cochrane Database of Systematic Reviews* 2000, Issue 4. [DOI: 10.1002/14651858.CD001088]

Marder 2004

Marder SR, Essock SM, Miller AL, Buchanan RW, Casey DE, Davis JM, Kane JM, Lieberman JA, Schooler NR, Covell N, Stroup S, Weissman EM, Wirshing DA, Hall CS, Pogach L, Pi-Sunyer X, Bigger JT, Friedman A, Kleinberg D, Yevich SJ, Davis B, Shon S. Physical health monitoring of patients with schizophrenia. *American Journal of Psychiatry* 2004;**161**(8):1334–49.

Marshall 1996

Marshall M. Case management: a dubious practice: Underevaluated and ineffective, but now government policy. *British Medical Journal* 1996;**312**:523–4.

Marshall 1998

Marshall M, Gray A, Lockwood A, Green R. Case management for people with severe mental disorders. *Cochrane Database of Systematic Reviews* 1998, Issue 2. [DOI: 10.1002/14651858.CD000050]

Marshall 2000

Marshall M, Lockwood A, Bradley C, Adams C, Joy C, Fenton M. Unpublished rating scales: a major source of bias in randomised controlled trials of treatments for schizophrenia. *British Journal of Psychiatry* 2000;**176**: 249–52. [PUBMED: 10755072]

Newman 1991

Newman SC, Bland RC. Mortality in a cohort of patients with schizophrenia: a record linkage study. *Canadian Journal of Psychiatry* 1991;**36**(4):239–45. [PUBMED: 1868416]

NICE 2006

National Collaborating Centre for Mental Health. The management of bipolar disorder in adults, children and adolescents, in primary and secondary care. National Clinical Practice Guideline Number 38. The British Psychological Society and Gaskell, 2006.

NIMH 1987

National Institute of Mental Health. Towards a model for a comprehensive community-based mental health system. National Institute of Mental Health Washington, DC: NIMH. 1987.

Overall 1962

Overall JE, Gorham DR. The Brief Psychiatric Rating Scale. *Psychological Reports* 1962;**10**:799–812.

Paton 2004

Paton C, Esop R, Young C, Taylor D. Obesity, dyslipidaemias and smoking in an inpatient population treated with antipsychotic drugs. *Acta Psychiatrica*

Scandinavica 2004;**110**(4):299–305. [PUBMED: 15352932]

RCPsych 2009

Phelan M. Physical health standards: the West London Mental Health Trust. Physical Health in Mental Health: Occasional Paper 67. Royal College of Psychiatrists, 2009: 21–23.

RCPsych 2009a

Osbourne D, Phelan M. Psychotropic prescribing. Physical Health in Mental Health:Occasional Paper 67. Royal College of Psychiatrists, 2009:55–61.

Robson 2007

Robson D, Gray R. Serious mental illness and physical health problems: a discussion paper. *International Journal of Nursing Studies* 2007;44(3):457–66. [PUBMED: 17007859]

Ruggeri 2000

Ruggeri M, Leese M, Thornicroft G, Bisoffi G, Tansella M. Definition and prevalence of severe and persistent mental illness. *British Journal of Psychiatry* 2000;**177**:149–55. [PUBMED: 11026955]

Schinnar 1990

Schinnar AP, Rothbard AB, Kanter R, Jung YS. An empirical literature review of definitions of severe and persistent mental illness. *American Journal of Psychiatry* 1990;**147**(12): 1602–8. [PUBMED: 2244636]

Tanenbaum 2005

Tanenbaum SJ. Evidence-based practice as mental health policy: Three controversies and a caveat. *Health Affairs* 2005;**24**(1):163–73.

Taylor 2009

Taylor D, Paton C, Kapur S. *The Maudsley prescribing guidelines*. 10th Edition. Informa Healthcare, 2009.

Tosh 2010

Tosh G, Gray R, Mala S, Clifton A, Bachner M. General physical health advice for people with serious mental illness. *Cochrane Database of Systematic Reviews* 2010, Issue 2. [DOI: 10.1002/14651858.CD008567.pub2]

Ukoumunne 1999

Ukoumunne OC, Gulliford MC, Chinn S, Sterne JAC, Burney PGJ. Methods for evaluating area-wide and organistation-based intervention in health and health care: a systematic review. *Health Technology Assessment* 1999;**3**(5): 1–75.

Weinmann 2009

Weinmann S, Read J, Aderhold V. Influence of antipsychotics on mortality in schizophrenia: systematic review. *Schizophrenia research* 2009;**113**(1):1–11. [PUBMED: 19524406]

White 2009

White J, Gray R, Jones M. The development of the serious mental illness physical health improvement profile. *Journal of Psychiatric and Mental Health Nursing* 2009;**16**(5):493–8. [PUBMED: 19538607]

WHO 1948

World Health Organization. WHO definition of Health [Preamble to the Constitution of the World Health Organization as adopted by the International Health Conference, New York]. Official records of the World Health Organization. New York, 1948; Vol. 2:100.

Xia 2009

Xia J, Adams CE, Bhagat N, Bhagat V, Bhoopathi P, El-Sayeh H, Pinfold V, Takriti Y. Loss to outcomes stakeholder survey: the LOSS study. *Psychiatric Bulletin* 2009;**33**(7): 254–7.

^{*} Indicates the major publication for the study

CHARACTERISTICS OF STUDIES

Characteristics of excluded studies [ordered by study ID]

| Study | Reason for exclusion |
|--------------|--|
| Jürgens 2008 | Allocation: randomised. Participants: people diagnosed with schizophrenia. Intervention: genotype monitoring versus intense clinical monitoring, not focusing on physical health |
| Lan 2007 | Allocation: randomised. Participants: people diagnosed with schizophrenia, schizoaffective disorder, or schizophreniform disorder. Intervention: monitoring of the effect of aripiprazole and aripiprazole plus haloperidol on prolactin levels, not focusing on physical health |
| Lan 2008 | Allocation: randomised. Participants: people diagnosed with schizophrenia, schizoaffective disorder, or schizophreniform disorder. Intervention: monitoring of the effect of aripiprazole and aripiprazole plus risperidone on prolactin levels, not focusing on physical health |
| Rostow 1980 | Allocation: randomised. Participants: people with "compulsive or persistent pacing", not necessarily having a diagnosis of serious mental illness |

Characteristics of studies awaiting assessment [ordered by study ID]

AstraZeneca 2006

| Methods | |
|-----------------------|-----------------|
| Participants | |
| Interventions | |
| Outcomes | |
| Notes | To be assessed. |
| | |
| Baptista 2010 | |
| Baptista 2010 Methods | |
| _ | |
| Methods | |

Baptista 2010 (Continued)

| Notes | To be assessed. |
|---------------|-----------------|
| Becker 2005 | |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |
| Notes | To be assessed. |
| Bobo 2011 | |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |
| Notes | To be assessed. |
| Bushe 2010 | |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |
| Notes | To be assessed. |
| Byerly 2010 | |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |

Byerly 2010 (Continued)

| Notes | To be assessed. |
|---------------|-----------------|
| Carmeli 2012 | |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |
| Notes | To be assessed. |
| ChiCTR-TRC | -12002273 |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |
| Notes | To be assessed. |
| Cordes 2011 | |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |
| Notes | To be assessed. |
| Fernandez 201 | 0 |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |

Fernandez 2010 (Continued)

| Notes | To be assessed. |
|---------------|-----------------|
| Henderson 20 | 11 |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |
| Notes | To be assessed. |
| IRCT2011071 | 87049N1 |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |
| Notes | To be assessed. |
| ISRCTN63382 | 2258 |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |
| Notes | To be assessed. |
| Kent 2011 | |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |

Kent 2011 (Continued)

| Notes | To be assessed. |
|----------------|-----------------|
| Kluge 2012 | |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |
| Notes | To be assessed. |
| Krakowski 20 | 11 |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |
| Notes | To be assessed. |
| Kreyenbuhl 20 | 011 |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |
| Notes | To be assessed. |
| L'Italien 2006 | |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |

L'Italien 2006 (Continued)

| Notes | To be assessed. |
|---------------|-----------------|
| Lencz 2010 | |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |
| Notes | To be assessed. |
| Lieberman 201 | 0 |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |
| Notes | To be assessed. |
| Loza 2011 | |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |
| Notes | To be assessed. |
| Miceli 2010 | |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |

Miceli 2010 (Continued)

| Notes | To be assessed. |
|---------------|-----------------|
| NCT00418171 | |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |
| Notes | To be assessed. |
| NCT00484302 | |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |
| Notes | To be assessed. |
| NCT01075295 | 5 |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |
| Notes | To be assessed. |
| NCT01368406 | 5 |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |

NCT01368406 (Continued)

| Notes | To be assessed. |
|---------------|-----------------|
| NCT01567124 | 4 |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |
| Notes | To be assessed. |
| NCT01606436 | 5 |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |
| Notes | To be assessed. |
| Newcomer 201 | 10 |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |
| Notes | To be assessed. |
| Nicol 2011 | |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |

Nicol 2011 (Continued)

| Notes | To be assessed. |
|---------------|-----------------|
| Nielsen 2012 | |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |
| Notes | To be assessed. |
| Ozguven 2011 | |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |
| Notes | To be assessed. |
| Peuskens 2007 | ' b |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |
| Notes | To be assessed. |
| Saddichha 201 | 1 |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |

Saddichha 2011 (Continued)

| Notes | To be assessed. |
|----------------|-----------------|
| Scheewe 2011 | |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |
| Notes | To be assessed. |
| Stroup 2011 | |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |
| Notes | To be assessed. |
| Tanasiewicz 20 | 011 |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |
| Notes | To be assessed. |
| Tessier 2010 | |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |

Tessier 2010 (Continued)

| Notes | To be assessed. |
|-----------------|-----------------|
| Tiwari 2010 | |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |
| Notes | To be assessed. |
| 何淑芬 2011 | |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |
| Notes | To be assessed. |
| 侯雪玲 2009 | |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |
| Notes | To be assessed. |
| 冯亚芬 2011 | |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |

冯亚芬 **2011** (Continued)

| Notes | To be assessed. |
|---------------|-----------------|
| 2010 | |
| 刘根 2010 | |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |
| Notes | To be assessed. |
| | |
| 刘清连 2009 | |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |
| Notes | To be assessed. |
| | |
| 张敏 2010 | |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |
| Notes | To be assessed. |
| | |
| 张炜 2009 | |
| Methods | |
| Participants | |
| Interventions | |

张炜 2009 (Continued)

| Outcomes | |
|------------------------|-----------------|
| Notes | To be assessed. |
| 影 素 2009 | |
| 影点 2009 | |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |
| Notes | To be assessed. |
| | |
| 徐清芝 2010 | |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |
| Notes | To be assessed. |
| | |
| 方崇芳 2011 | |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |
| Notes | To be assessed. |

| 李玉凤 2010 | |
|-----------------|-----------------|
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |
| Notes | To be assessed. |
| 李红 2010 | |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |
| Notes | To be assessed. |
| 李萍 2010 | |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |
| Notes | To be assessed. |
| 林素兰 2010 | |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |
| Notes | To be assessed. |

| 温广妹 2009 | |
|-----------------|-----------------|
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |
| Notes | To be assessed. |
| 王宪琴 2010 | |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |
| Notes | To be assessed. |
| 王继红 2010 | |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |
| Notes | To be assessed. |
| 廖世铭 2010 | |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |
| Notes | To be assessed. |

| 苏勉 2010 | |
|-----------------|-----------------|
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |
| Notes | To be assessed. |
| 苏爱东 2010 | |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |
| Notes | To be assessed. |
| 蔡连秀 2010 | |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |
| Notes | To be assessed. |
| 爾瑞云 2010 | |
| Methods | |
| Participants | |
| Interventions | |
| Outcomes | |
| Notes | To be assessed. |

DATA AND ANALYSES

This review has no analyses.

ADDITIONAL TABLES

Table 1. Suggested design of study

| Methods | Allocation: randomised, clearly described. Blinding: single - particular to specific outcomes (see below). Duration: 6 months. |
|---------------|--|
| Participants | Diagnosis: schizophrenia, or any serious mental illness. N=450.* Age: any. Sex: both. History: any. |
| Interventions | General physical health care checklist (e.g. Physical Health Improvement Profile see White 2009): administered by Care Co-ordinator. N=150. Specific aspect of physical health care checklist (e.g. BSDH 2000): administered by Care Co-ordinator. N=150. Standard care: administered by Care Co-ordinator. N=150. |
| Outcomes | Death. Morbidity: serious/minor, categorised by type, rates of events - general or specific. Healthy days. Service use: visit to heath care practitioner. Acceptability of checklist. Compliance: with physical health care advice, including treatments. Adverse effects: any. |
| Notes | * For 20% difference between groups for a binary outcome to be highlighted with reasonable degree of confidence 150 people are needed per group |

WHAT'S NEW

Last assessed as up-to-date: 18 January 2010.

| Date | Event | Description |
|-----------------|---------|---|
| 17 October 2012 | Amended | Update search of Cochrane Schizophrenia Group's Trial Register (see Search methods for identification of studies), 61 studies added to awaiting classification. |

HISTORY

Protocol first published: Issue 1, 2010 Review first published: Issue 3, 2010

| Date | Event | Description |
|---------------|---------|---|
| 17 March 2010 | Amended | Amendment of outcomes to be included in summary of findings table |
| 17 March 2010 | Amended | Previously combined studies split into separated studies and included in Excluded Studies section |

CONTRIBUTIONS OF AUTHORS

Graeme Tosh - project initiation, protocol writing, primary reviewer, results and discussion writing.

Andrew Clifton - co-reviewer and liaison on discussion and results writing.

Shereen Mala - co-reviewer, liaison on protocol.

Mick Bachner - co-reviewer, screened results of electronic search.

DECLARATIONS OF INTEREST

None known.

SOURCES OF SUPPORT

Internal sources

• University of Nottingham, UK.

External sources

• National Institute for Health Research (CLAHRC), UK.

DIFFERENCES BETWEEN PROTOCOL AND REVIEW

Subsequent to the publication of the protocol we identified that our search strategy may have excluded some relevant studies so we added the words 'advice' and 'monitor' in the title of references section and added *advice* and *monitor* to the interventions of study field.

The original search phrase: [(*physical* or *cardio* or *metabolic* or *weight* or *HIV* or *AIDS* or *Tobacc* or *Smok* or *sex* or *medical* or *dental* or *alcohol* or *oral* or *vision* or *sight*or *hearing* or *nutrition* in title of REFERENCES) AND (*education* OR *health promot* OR *preventi* OR *motivate* in interventions of STUDY)] yielded 2326 references whilst the new search phrase: [(*physical* or *cardio* or *metabolic* or *weight* or *HIV* or *AIDS* or *Tobacc* or *Smok* or *sex* or *medical* or *dental* or *alcohol* or *oral* or *vision* or *sight*or *hearing* or *nutrition* or *advice* or *monitor* in title of REFERENCES) AND (*education* OR *health promot* OR *preventi* OR *motivate* or *advice* or *monitor* in interventions of STUDY)] yielded 2383 references.

INDEX TERMS

Medical Subject Headings (MeSH)

*Health Status; *Quality of Life; Disease Progression; Mental Disorders [*complications]

MeSH check words

Humans